Author's response to reviews

Title: Evaluation of a toolkit to improve cardiovascular disease screening and treatment for people with type 2 diabetes: protocol for a cluster-randomized pragmatic trial

Authors:

Baiju R Shah (baiju.shah@ices.on.ca)
Onil Bhattacharyya (bhattacharyyao@smh.toronto.on.ca)
Catherine Yu (yuca@smh.toronto.on.ca)
Muhammad Mamdani (mamdanim@smh.toronto.on.ca)
Janet A Parsons (parsonsj@smh.toronto.on.ca)
Sharon E Straus (sharon.straus@utoronto.ca)
Merrick Zwarenstein (merrick.zwarenstein@ices.on.ca)

Version: 4 Date: 21 April 2010

Author's response to reviews:

The only revisions relate to changes in the timing of the distribution of the Toolkit to the control arm of the study, which has evolved since the submission of the original manuscript. Instead of receiving it in Winter 2010, they will now receive it in Spring 2010. This change has been reflected in the abstract (under design) and on page 4, section "Intervention".

As a result of this change, the outcome measures will now be determined until March 2010, instead of February 2010. This change has been reflected in the abstract (under design) and on page 5, sections "Study participants" and "Data collection and outcomes".